Author's response to reviews

Title: Involvement of males in antenatal care, birth preparedness, exclusive breast feeding and immunizations for children in Kathmandu, Nepal

Authors:

Dharma N Bhatta Mr. (dnbhatta@yahoo.com)

Version: 5 Date: 7 January 2013

Author's response to reviews: see over
To

The Chief Editor
BMC Pregnancy and Childbirth

Subject: Revision of submitted manuscript.

Dear Editor-in-Chief,

Please find enclosed my revised manuscript, “Involvement of male in antenatal care, birth preparedness, exclusive Breast feeding and immunization for child in Kathmandu, Nepal” by Dharma Nand Bhatta, which was submitted for publication as a peer review in BMC pregnancy and childbirth. The following point-by-point revisions have been made in the manuscript based on reviewers’ comments.

- I have been used the NPR instead of NRs. in the manuscript. Exchange rate to USD and NPR has been included in abstract and text.

- I have been changed other than Hindu religion instead of non-Hindu.

- Structure of the last sentence on ORs and CIs has been corrected.

- References have been corrected according to journal style.

- Table 1: Space has been corrected.

- Alphabets have been used on the footnotes of tables instead of characteristics.

I look forward to hearing from you at your earliest convenience.

Sincerely yours’

Dharma Nand Bhatta
dnbhatta@yahoo.com